Author’s response to reviews

Title: Improving interMediAte RisK management. MARK study.

Authors:

Ruth Martí (rmarti.girona.ics@gencat.cat)
Didac Parramon (didacparramon@gmail.com)
Luís García-Ortiz (Lgarciao@usal.es)
Fernando Rigo (frigoc5@gmail.com)
Manuel A Gómez-Marcos (magomez@usal.es)
Irene Sempere (isempere@ibsalut.caib.es)
Natividad García-Regalado (regarna@gmail.com)
Jose I Recio-Rodriguez (donrecio@gmail.com)
Cristina Agudo-Conde (cagudoconde@yahoo.es)
Natalia Feuerbach (nfeuerbach@gmail.com)
Maria Garcia-Gil (mgarcagiil.girona.ics@gencat.cat)
Anna Ponjoan (aponjoan@idiapjgol.info)
Miquel Quesada (mquesada.girona.ics@gencat.cat)
Rafel Ramos (rramos.girona.ics@gencat.cat)

Version: 4 Date: 11 October 2011

Author’s response to reviews:

Dear editorial board of BMC Cardiovascular Disorders

Thank you for your interest and your suggestions and comments to our manuscript

Description of the changes:

Major revisions:

1-I corrected the list of authors in the submission system and I re-wrote them as they are in the manuscript. Some last names are written with a hyphenated compound name, this is because they want their second surname appears in the list of authors.

Minor revisions:

1-I changed Methods and design to Methods/Design at the Abstract

Sincerely yours,

Dídac Parramon on behalf of the authors